

THE UNIVERSITY of EDINBURGH

Edinburgh Research Explorer

Subglacial Freshwater Drainage Increases Simulated Basal Melt of the Totten Ice Shelf

Citation for published version:

Gwyther, DE, Dow, CF, Jendersie, S, Gourmelen, N & Galton-fenzi, BK 2023, 'Subglacial Freshwater Drainage Increases Simulated Basal Melt of the Totten Ice Shelf', Geophysical Research Letters, vol. 50, no. 12, e2023GL103765. https://doi.org/10.1029/2023GL103765

Digital Object Identifier (DOI):

10.1029/2023GL103765

Link: Link to publication record in Edinburgh Research Explorer

Document Version: Publisher's PDF, also known as Version of record

Published In: Geophysical Research Letters

Publisher Rights Statement:

© 2023. The Authors

General rights

Copyright for the publications made accessible via the Edinburgh Research Explorer is retained by the author(s) and / or other copyright owners and it is a condition of accessing these publications that users recognise and abide by the legal requirements associated with these rights.

Take down policy The University of Edinburgh has made every reasonable effort to ensure that Edinburgh Research Explorer content complies with UK legislation. If you believe that the public display of this file breaches copyright please contact openaccess@ed.ac.uk providing details, and we will remove access to the work immediately and investigate your claim.



ŗ



Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2023GL103765

Key Points:

- Subglacial discharge has a strong local effect on melting beneath the Totten Ice Shelf
- Addition of buoyancy explains the change in melting
- Discharge has far-reaching impacts due to the extent of the buoyant plume

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

D. E. Gwyther, david.gwyther@gmail.com

Citation:

Gwyther, D. E., Dow, C. F., Jendersie, S., Gourmelen, N., & Galton-Fenzi, B. K. (2023). Subglacial freshwater drainage increases simulated basal melt of the Totten Ice Shelf. *Geophysical Research Letters*, *50*, e2023GL103765. https://doi. org/10.1029/2023GL103765

Received 21 MAR 2023 Accepted 31 MAY 2023

Author Contributions:

Conceptualization: David E. Gwyther, Christine F. Dow, Benjamin K. Galton-Fenzi Data curation: David E. Gwyther, Christine F. Dow Formal analysis: David E. Gwyther Funding acquisition: Benjamin K. Galton-Fenzi Investigation: Christine F. Dow, Stefan Jendersie, Noel Gourmelen, Benjamin K. Galton-Fenzi Methodology: David E. Gwyther, Christine F. Dow, Stefan Jendersie, Benjamin K. Galton-Fenzi Project Administration: David E. Gwyther Software: David E. Gwyther, Christine F. Dow, Stefan Jendersie, Noel Gourmelen Validation: David E. Gwyther

© 2023. The Authors. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is

Subglacial Freshwater Drainage Increases Simulated Basal Melt of the Totten Ice Shelf

David E. Gwyther¹, Christine F. Dow², Stefan Jendersie³, Noel Gourmelen⁴, and Benjamin K. Galton-Fenzi^{5,6}

¹School of Earth and Environmental Sciences, The University of Queensland, St Lucia, QLD, Australia, ²Department of Geography and Environmental Management, University of Waterloo, Waterloo, ON, Canada, ³Antarctic Research Centre, Victoria University of Wellington, Wellington, New Zealand, ⁴School of Geosciences, University of Edinburgh, Edinburgh, UK, ⁵Australian Antarctic Division, Kingston, TAS, Australia, ⁶Australian Antarctic Program Partnership, Institute for Marine and Antarctic Studies, University of Tasmania, Hobart, TAS, Australia

Abstract Subglacial freshwater discharge from beneath Antarctic glaciers likely has a strong impact on ice shelf basal melting. However, the difficulty in directly observing subglacial flow highlights the importance of modeling these processes. We use an ocean model of the Totten Ice Shelf cavity into which we inject subglacial discharge derived from a hydrology model applied to Aurora Subglacial Basin. Our results show (a) discharge increases melting in the vicinity of the outflow region, which correlates with features observed in surface elevation maps and satellite-derived melt maps, with implications for ice shelf stability; (b) the change in melting is driven by the formation of a buoyant plume rather than the addition of heat; and (c) the buoyant plume originating from subglacial discharge-driven melting is far-reaching. Basal melting induced by subglacial hydrology is thus important for ice shelf stability, but is absent from almost all ice-ocean models.

Plain Language Summary Grounding line subglacial discharge is the outflow of freshwater at the region where Antarctic ice shelves begin to float. This process is difficult to observe as the outflow region is typically below ice up to several kilometers thick. Using a computer model, we show how the outflow of this fresh water melts the underside of an Antarctic ice shelf. We demonstrate that the outflow leads to a strong increase in melting, which is driven not by the heat in the outflow, but rather by the low salinity water's buoyancy. This buoyancy leads to the formation of a plume that rapidly ascends the underside of the ice shelf and can carry even warmer water to the ice, leading to increased melting. Given that melting of ice shelves can destabilize the grounded ice upstream of the floating region, we show that the relationship between subglacial discharge and ice shelf melt is important for understanding the stability of Antarctic ice.

1. Introduction

Thinning of ice shelves currently occurs primarily at the ice-ocean boundary, with high melting from the flux of relatively warm water into ice shelf cavities (e.g., Davis et al., 2023), which can lead to glacier acceleration (Dupont & Alley, 2005; Scambos et al., 2004) and reduction in ice sheet stability (Fürst et al., 2016).

A factor that has been only minimally considered is the role of subglacial freshwater (SGFW), which flows into Antarctic ice shelf cavities from beneath the ice sheet (Dow et al., 2020). As subglacial water is a source of highly buoyant water compared to the ambient ocean conditions, it could play an important role in ice shelf basal melting (Hewitt, 2020). Subglacial outflow has been connected with the formation of basal channels in ice shelves; these channels can be several kilometers wide and hundreds of meters high to the extent that they are observable in satellite optical imagery (Alley et al., 2016; Gourmelen et al., 2017; Le Brocq et al., 2013; Marsh et al., 2016). Examination of subglacial outflow locations compared with ice shelf melt rates at Getz Ice Shelf, West Antarctica (Wei et al., 2020) and the Weddell Sea Region (Dow et al., 2022) also showed a close link between the channel location, SGFW discharge volume, and rates of adjacent ice shelf melt.

However, to date, there are only extremely limited in situ observations of the subglacial environment and drainage, for example, Davis et al. (2023), who observed subglacial discharge below Thwaites Ice Shelf. Furthermore, the role of subglacial hydrology in frontal melting of tidewater glaciers in Greenland (Andrés et al., 2020; Cai et al., 2017; S. J. Cook et al., 2020; Fried et al., 2015; Gladish et al., 2012; Slater et al., 2015) along with

properly cited.



Visualization: David E. Gwyther, Christine F. Dow Writing – original draft: David E. Gwyther, Christine F. Dow, Stefan Jendersie, Benjamin K. Galton-Fenzi Writing – review & editing: David E. Gwyther, Christine F. Dow, Stefan Jendersie, Noel Gourmelen, Benjamin K. Galton-Fenzi alterations in larger-scale fjord circulation (Slater et al., 2018) demonstrate that the impact of subglacial hydrology can be a significant component of ice-ocean interactions.

Motyka et al. (2003) proposed that subglacial discharge drives strong buoyant convection, which forms an ascending plume, entrains warmer waters and drives ice front melting. Jenkins (2011) extended this concept for Antarctica, by unifying the concept of convection-driven melting (where buoyancy generates a plume that subsequently drives melting) with the commonly understood melt driven-convection paradigm (where melting releases a buoyant plume; e.g., the ice pump mechanism, Lewis & Perkin, 1986).

To date, the only model to simulate the input of Antarctic subglacial discharge into a realistic ocean model is Nakayama et al. (2021). SGFW discharge was directed into the ice shelf cavity from topographical routing estimates (as opposed to direct hydrological modeling) and included at two main output locations below Pine Island Ice Shelf in a high resolution ice shelf-ocean model. They showed substantial melt increases in the vicinity of drainage output and suggested that subglacial drainage is necessary to match satellite melt estimates (Nakayama et al., 2021).

In our study, we combine outputs from a subglacial hydrology model into an ice shelf-ocean model to assess not only how the steady-state freshwater flux impacts ice shelf melting, but also why melting is impacted.

2. Methods

2.1. Study Area

The Totten Glacier and adjoining Totten Ice Shelf (TIS) are the most rapidly changing region of the East Antarctic (Smith et al., 2020) and currently hold back ~3.5 m of sea level rise potential from the Aurora Subglacial Basin (Greenbaum et al., 2015). Observations of this region suggest that it is undergoing ice surface lowering, retreat and mass loss (e.g., Flament & Rémy, 2012; Li et al., 2015; Pritchard et al., 2012; Roberts et al., 2018; Velicogna et al., 2020), although complex variability (Adusumilli et al., 2020; Gwyther et al., 2018) limits the ability to detect and attribute recent change. The observed oceanographic regime (Greene, Blankenship, et al., 2017; Rintoul et al., 2016; Silvano et al., 2017, 2019; T. P. Tamura et al., 2022; Williams et al., 2011) and bathymetric connections (e.g., Greenbaum et al., 2015) beneath and adjacent to the TIS indicate the potential for warm, off-shelf waters to access the cavity and drive strong basal melting.

Previous modeling (e.g., Gwyther et al., 2014; Van Achter et al., 2022; Xia et al., 2023) indicated that inflow into the cavity (Figures 1a and 1b) is on the eastern side and can carry relatively warm water (see Figure S1 in Supporting Information S1). Ice shelf meltwater flows out of the cavity adjacent to the western grounding line as a topographically constrained boundary current, before exiting the cavity.

It is unknown how subglacial discharge, which drains into the cavity in relatively high volumes (Dow et al., 2020), impacts ice shelf melting and hence stability in this sensitive region.

2.2. GlaDS

The Glacier Drainage System (GlaDS) model is a 2D finite element model of subglacial hydrology (Werder et al., 2013). It includes a distributed system that develops on the elements where water depth, pressure and flux are calculated. Channels develop along element edges and form naturally as result of the pressure and flux conditions in the adjacent distributed elements. The application of GlaDS to the Aurora Subglacial Basin feeding into the TIS is described in Dow et al. (2020). Inputs include ice surface and bed topography interpolated from airborne radar lines, along with ice basal velocity and basal melt rates calculated from the Ice Sheet System Model. Flow routing in both the distributed and channelized system, including the location of outflow over the grounding line, is driven by a combination of the topography and the variability of water pressure in the system. The model outputs are from runs that have reached steady state.

Three sensitivity experiments were conducted with a range of flux conditions into the TIS cavity. As the total discharge volume did not vary greatly between experiments, the resultant impact on melt rates is low and we present only the medium conductivity results here (see Figure S2 in Supporting Information S1). The maximum inflow is from a channel at the head of the ice shelf (Figure 1c), with a peak output of 25 m³ s⁻¹, contributing \sim 78% of the total SGFW inflow.





Figure 1. Ocean model (a) bathymetry and (b) ice draft are shown. The contour interval is 1,000 m. (c) Modeled locations of drainage channels (black lines) with at least $0.5 \text{ m}^3 \text{ s}^{-1}$ discharge are shown. Filled and area-scaled circles show discharge rate on a logarithmic color scale. TIS surface elevation is from REMA (Howat et al., 2019). The background image is the MEaSURES MODIS Antarctic mosaic (Scambos et al., 2007).

2.3. ROMS

We use the Regional Ocean Modeling System (ROMS; Shchepetkin & McWilliams, 2005) with ice-ocean thermodynamics and mechanical pressure (Dinniman et al., 2007; Galton-Fenzi et al., 2012). Heat and salt exchange across the ice-ocean boundary layer is captured through the three-equation parameterization (Hellmer & Olbers, 1989; Holland & Jenkins, 1999) with velocity-dependent exchange coefficients (McPhee et al., 1987) and a quadratic drag formulation ($C_D = 0.003$). Vertical mixing is parameterized with the K-profile parameterization (Large et al., 1994). The model has 31 sigma-coordinate layers, with approximately 2.5 km horizontal resolution at the ice shelf. The model domain is the same as in Gwyther et al. (2018), and includes the bathymetry estimated from gravity inversion (Greenbaum et al., 2015) and ice draft from airborne radar soundings (Figures 1a and 1b). The experimental conditions include tidal forcing (following Galton-Fenzi et al., 2012), repeat 1992–2017 climatological forcing of the surface and lateral boundaries, with a spin-up period of 20 years. Lateral boundary conditions (velocity, temperature and salinity) are from ECCO2 (Menemenlis et al., 2008); surface boundary conditions include wind forcing (ERA-interim; Dee et al., 2011) and buoyancy fluxes (T. Tamura et al., 2016) to parameterize sea ice formation.

The existing river source module within ROMS (e.g., Zhang et al., 2009) was employed to simulate subglacial discharge. There is an addition to the horizontal momentum term (at the rate described by the river source flux), with associated tracer fluxes diverged by the advection operator, at each cell face identified as a channel outlet. As there is no extra water volume input, it should be thought of as a modification to the existing conditions in the cell. Input was vertically distributed in a linearly increasing manner toward the top of the column. This is to ensure that SGFW entering the model is not buoyantly unstable, which could lead to numerical instabilities. This represents a balance between the realistic entry location (likely at the narrow grounding zone located just above the seabed) and model requirements (where the generation of unstable stratification can lead to numerical instability). Discharge is set to enter at the freshwater pressure melting point with zero salinity.

2.4. Satellite Melt Rates

Melt rates are inferred with the mass conservation approach constrained using satellite data following Gourmelen et al. (2017). This method combines satellite observations and climate models balancing mass loss from basal melting, surface lowering through volume divergence, with mass gain from surface processes. Lagrangian surface elevation change is generated from CryoSat-2 swath radar altimetry from 2010 to 2020 (Gourmelen et al., 2018)



Geophysical Research Letters



Figure 2. Time-mean melt rate beneath the TIS is shown for (a) the satellite-inferred method for the period 2010–2020, and (b) ocean model simulation including the influence of SGFW. In (c), the percentage difference in time-mean melt rate between the experiments with SGFW and without SGFW is shown. In (a) melt rates have been removed in regions where heavy crevassing impedes the melt rate algorithm (e.g., at the calving front). In (b and c), the model calving front and grounding line are delineated with orange lines. Background image is from the MODIS Mosaic Of Antarctica (Scambos et al., 2007). Locations mentioned in text are marked in panel (a), as well as the north direction. The TI05 Autonomous Phase-sensitive Radar Echo Sounding site (Vaňková et al., 2021) is marked with a circle on the same color scale in (a and b). *X* and *Y* coordinates are on a polar stereographic projection (standard latitude of 71°S).

and surface ice velocity is from the ITS_LIVE data set (Gardner et al., 2018), with a full description given in Gourmelen et al. (2017). Surface mass balance is obtained from the Regional Atmospheric Climate Model version 2 (RACMO2; Wessem et al., 2018).

2.5. Experiments

To assess the impact of SGFW on melt rates and distribution, a series of experiments are performed with ROMS. All experiments are conducted as sensitivity studies from the end of the spin-up phase, and consist of a single, perturbed year as ocean state evolution beyond this period limits our ability to perform comparative analyses. This study consists of two sets of experiments.

The first set compares a "no-flow" control scenario (addition of no SGFW) to a sensitivity run with the addition of SGFW ("with SGFW"; see GlaDS description and Supporting Information S1).

The second set of experiments compares two further model runs with subglacial discharge to the control experiment (no-flow), but the input SGFW flux is switched between having only the temperature tracer modified ("temperature-only" experiment) or only the salt tracer modified ("salt-only experiment"). The other tracer is unmodified.

3. Results

3.1. Mean Melt Rates

Melt rates estimated with the satellite-inferred method (Figure 2a) show highest melt through the western ice shelf and grounding zone and weaker to moderate melt over much of the rest of the shelf. This leads to a TIS-wide area-averaged melt of 12 ± 1.4 m yr⁻¹ or a total melt mass flux of 64 ± 7.4 Gt yr⁻¹. Autonomous Phase-sensitive Radar Echo Sounding (ApRES) observations taken on the southern grounding line region of the TIS in 2017/2018 (Figures 2a and 2b), showed melt rates of 22 ± 2 m yr⁻¹ (Vaňková et al., 2021).

Melt rates for ROMS with the SGFW flux (Figure 2b) are high (maximum of 50 m yr⁻¹) in the deeper grounding line regions and through the eastern channel region, while melting across the northern, shallower portions of the ice shelf is generally less than ~5 m yr⁻¹. The TIS-wide area average melt rate is 8.3 m yr⁻¹ (a total melt mass flux of 54 Gt yr⁻¹).

The glaciological/satellite method shows higher melting than the ocean model in the deep grounding line region and on the western ice shelf flank; outside of these regions, both products compare relatively well. The simulated melt rate compares well with the observation of melt taken by radio echo sounding at that location (compare 21 m yr⁻¹ in the simulation and the observation of 22 ± 2 m yr⁻¹).

Multiple reasons exist for differences between modeled and observed melt rates. First, the model forcing epoch differs from the satellite-product time period. Ocean models are also reliant on accurate bathymetry (e.g., Goldberg et al., 2019), which is still unmeasured beneath most of the TIS. Furthermore, horizontal resolution, while relatively high in this simulation, may not capture small-scale flow features (Dinniman et al., 2016). Like other ice shelf-ocean models, sub-grid-scale parameterisations are used for capturing many unresolved processes—however, these are likely a source of error. Finally, while we have included SGFW discharge, errors in geothermal heating may lead to an underestimation of subglacial fluxes. Sources of error in the satellite-derived product include from the surface accumulation model, firn compaction, and vertical strain profile (e.g., S. Cook et al., 2023; Vaňková & Nicholls, 2022). Nevertheless, the modeled melt rates display a similar pattern to satellite-inferred melt rates, and agree with the Autonomous Phase-sensitive Radar Echo Sounding observations, making them suitable for exploring the impact of subglacial drainage on melt rate.

3.2. Exploring the Addition of SGFW

The comparative impact of subglacial drainage on ice shelf basal melting is small overall but strong in localized regions (Figure 2c). The percentage difference in melt rates (where a positive value indicates higher melting with SGFW) shows that while the area-average increase in melting is $\sim 3.3\%$ (an increase in mass flux of 1.38 Gt yr⁻¹), there are strong, local variations due to the input of subglacial water. Change in melting is generally strongest beneath the deepest, thickest sections of the ice shelf (adjacent to the grounding line; Figure 2c) where discharge increases melting by 20%–30% (with peak increases of $\sim 50\%$). For example, for the region of the ice shelf where the UTM northings coordinate is greater than -1,030 km, melting increases on average by 29%. This results in a total increase in melt mass flux of 1.0 Gt yr⁻¹ over a relatively small 650 km². Enhanced melting also occurs along the northwestern side of the ice shelf (in the location of the outflowing topographically-constrained boundary current) and in the eastern channel (Figure 2c). This can be compared with a broader region along the southeastern (inflow) edge of the ice shelf where melt rates decrease by 2%–3% compared to the no-flow run (Figure 2c). The mean change in melt for all cells exhibiting an increase is 4.9\%, while the mean percentage decrease in melt for all cells with decreased melt is 1.6\%.

The reduced melting on the eastern side of the ice shelf results from cooler waters at the ice/ocean boundary (Figures S2c and S2d in Supporting Information S1). This occurs due to the increased presence of cold meltwater (Figures S3a and S3b in Supporting Information S1) and increased inflow of colder surface waters from outside the cavity. Increased inflow of warmer water occurs at depth (Figure S4 in Supporting Information S1), but this flows to the grounding line with minimal contact with the intermediate waters above (Figure S3c in Supporting Information S1).

The spatial distribution of subglacial discharge is shown with a passive tracer dye in Figure 3 (also see Movie S1). The SGFW dye enters at the grounding line (Figure 3a) and by 30 days has reached the calving front (Figure 3d). On leaving the cavity, the SGFW dye enters the Antarctic Coastal Current, where is advected westwards around Law Dome (Figures 3e-3g). Within 5 months (Figure 3h), the SGFW dye has traveled over 500 km.

3.3. Single Tracer Experiments

In Figure 4, the change in melting compared to the "no-flow" run is shown for the inflow of SGFW with only a modified temperature tracer or with only a modified salinity tracer. Enhanced melting is shown to be due to freshening from the SGFW, and with little to no impact from the addition of the heat in the SGFW (Figure 4a).

To better understand the dominance of freshening in the enhanced melting signature, we explore the factors driving the melt rate (Figure 4b). The two parameters that control basal melting are the thermal driving and the



Geophysical Research Letters

10.1029/2023GL103765



Figure 3. Depth-integrated SGFW dye concentration is shown relative to the initial concentration on a logarithmic scale. (a–d) Show the first 30 days of the year (DOY) and (e–h) show DOY 60 to 150. The calving front is delineated with an orange line. Background image is from the MODIS Mosaic Of Antarctica (Scambos et al., 2007).

friction velocity (Figures 4c and 4d). The thermal driving is defined here as $T_* = T_b - T_f$, where T_f is the in situ freezing point and T_b is the temperature at the ice shelf base; Figures 4c, 4f, and 4i. The friction velocity is defined as $u_* = \sqrt{(C_d)}|u|$, where C_d is the drag coefficient and u is velocity magnitude in the cell adjacent to the ice base; Figures 4d, 4g, and 4j. For the run with SGFW, melting (Figure 4b) is strongly influenced by the distribution of ocean heat (Figure 4c), as shown by the correlation between melting and T_* . The higher T_* and melting toward the grounding line results from the depression in the freezing point under thicker ice. The influence of ocean currents on melting is less obvious, although regions of increased u_* often coincide with increased melting, noticeably, for example, near constricted geometry or pinning points (Figure 4d).

The temperature-only scenario has minimal impact on melting (Figure 4e) compared to the control scenario (mean of -0.2%), with only some minor differences near to the ice front emerging in the thermal driving and velocities (Figures 4f and 4g). Given that this region is most proximate to the dynamic open ocean, it is not surprising that minor differences could evolve here between the control and sensitivity scenarios.

The salinity-only scenario, however, has a very strong impact on melting (mean of 3%), essentially entirely explaining the impact of SGFW on melt rates; for example, compare Figures 2c and 4h. Thermal driving is strongly modified by subglacial discharge (Figure 4i), even though no extra heat is being added. Therefore, this increased T_* must result from the increased buoyant circulation (see increased u_* in Figure 4j) entraining and supplying more ocean heat from deeper ocean waters to the base of the ice shelf. This demonstrates that the addition of fresh and thus buoyant water into the cavity enhances flow past the ice shelf and is the principal cause of the stronger melting.

4. Discussion

4.1. Subglacial Discharge Has a Strong Local Impact on Melting

The area-averaged increase in melting due to subglacial flux is relatively weak when examined over the whole TIS (around 3.3% or 1.4 Gt yr⁻¹ of extra ice lost). However, some of the increased mass flux is balanced by a reduction in melting in the "oceanic inflow" region of the ice shelf (Figure 2c), which results from altered cavity circulation and inflow of cooler surface waters.

For areas of increased melt there is a mean increase of 5%, with strong localized impacts of 25%-30% in the deep grounding line and eastern channel regions, where the discharge rate from subglacial channels are ~ 25 and





Figure 4. (a) Melt rate is shown for the no-flow (blue line), SGFW (orange line), temperature-only (yellow line) and salt-only (purple line) scenarios. The (b) melt rate, (c) thermal driving and (d) friction velocities are shown for the SGFW scenario. The % difference (relative to no-flow) is shown in (e) melt, (f) thermal driving and (g) friction velocity in the temperature-only scenario, and the same but for the (h–j) salt-only scenario.

1.4 m³ s⁻¹, respectively. This result agrees with Nakayama et al. (2021), who demonstrate very strong localized increases in melting near to subglacial outflow locations. The impact of these concentrated discharge plumes and corresponding high shelf melt rates are demonstrated by the presence of ice shelf channels. The shelf channels can be seen in satellite elevation products (e.g., the Reference Elevation Model of Antarctica Howat et al., 2019) and advect from the grounding line at the locations where subglacial discharge is greater than 1 m³ s⁻¹ toward the ice shelf terminus (Figure 1c). In addition, satellite-derived elevated ice shelf melt rates and corresponding grounding line retreat are also focussed in these regions of high subglacial discharge and high ROMS melt (Adusumilli et al., 2020; Flament & Rémy, 2012). Here our data indicate that the role of subglacial discharge is key for high melt in these sensitive areas. Ice loss at the grounding line (as opposed to in the middle of the ice shelf) will lead to more rapid grounding line retreat and, depending on the bedrock geometry, the potential initiation of an unstable retreat scenario (Marine Ice Sheet Instability; Tsai et al., 2015; Weertman, 1974).

4.2. Increased Melting Results From Freshening and Addition of Buoyancy

Nakayama et al. (2021) show higher melt rates are primarily explained by increased ocean currents, but suggest increased thermal driving could be due to the injection of subglacial discharge, which is warm relative to the seawater freezing point. Our experiments suggest that the thermal component of subglacial water has very little impact on ice shelf melt. Indeed, if subglacial discharge is at the pressure freezing point, then any mixture between it and ambient waters will likely result in reduced thermal driving, except potentially in the case of supercooled ambient waters. This is due to the resultant SGFW–ambient water mixture being closer to the freezing point line (as illustrated with warm and cold cavity conditions on a θ -S diagram with Gade lines (Wåhlin et al., 2010) in Figure S5 in Supporting Information S1).

Instead, regions of increased melting result from the increased buoyancy-driven plume (Motyka et al., 2003) and are independent of temperature. The role of subglacial water, at least in the vicinity of subglacial channel outlets, is therefore not to melt directly, but to rapidly initiate strong buoyant plumes, entrain warm ocean waters to the ice shelf boundary layer, and increase melting, as simulated by Jenkins (2011). Beyond the convection-driven melt zone, release of ice shelf meltwater (rather than subglacial discharge) is the primary contributor to buoyancy-driven melting.

The version of ROMS applied here makes the hydrostatic assumption and so cannot resolve non-hydrostatic dynamics. Nevertheless, if there is strong stratification and weaker flow speeds, then by the criterion of validity for the hydrostatic approximation (Marshall et al., 1997), the approximation we make in the vicinity of the channel outlet likely still holds. Furthermore, the dynamics of the buoyant circulation initiated by SGFW discharge, beyond the grounding line and at-or-greater than the grid resolution, are largely identical to other sub-ice shelf plumes, which have been simulated by many similar hydrostatic models (e.g., Nakayama et al., 2021).

4.3. Small Volumes With Broad Impacts

The volumes of subglacial water necessary to increase entrainment of ocean waters and hence basal melting, are small. Although the greatest increases in melting occur where the greatest freshwater discharge enters the cavity, there is also enhanced melt in regions where subglacial discharges are as low as 3 cm³ s⁻¹. Furthermore, the subsequent outflow plume originating from subglacial-driven melting, identified by the region of increased melt on the western flank of the ice shelf, is ~5 km wide and extends ~100 km in length, from the grounding line to the ice shelf terminus. This suggests that, given the widescale subglacial drainage around Antarctica (Le Brocq et al., 2013), enhanced melting from this process is extensive, with the strongest impact on melting occurring at the grounding line in areas adjacent to large subglacial outflow channels. As such, it is important that the impacts of subglacial drainage are included in ocean–ice shelf models to project ice shelf evolution in the Antarctic.

It is unclear at what stage future atmospheric conditions and drainage due to surface melt will begin to significantly influence the volumes of subglacial outflow in Antarctica. Furthermore, it is unclear what the impact of rapid and strong pulses of discharge would be, for example, from lake drainage (e.g., Fricker et al., 2007; Warner et al., 2021).

5. Conclusion

We find increases in TIS melting resulting from SGFW outflow. Increases are locally strong in regions near the grounding line, likely important for maintaining ice shelf stability. The formation of a strong buoyant plume,

19448007, 2023, 12, Downloaded

from

initiated by SGFW, ascends the sloping ice shelf base and subsequently increases melting. The spatial impact of discharge and increased melting is shown to reach far beyond the immediate grounding line zone.

Changes in ocean heat supply are likely the most important factor for ice shelf stability. However, to understand current and future melt rates, we need to account for the additional impact of convection-driven melting resulting from subglacial discharge. Should surface melting increase above present-day levels and this additional meltwater reaches the bedrock, the volume of subglacial discharge would likely be significantly more - and the impact on basal melting profoundly different.

Data Availability Statement

Ocean model output is available at https://doi.org/10.17605/OSF.IO/MBR9F, processing scripts are available at https://doi.org/10.5281/zenodo.5218955, and gridded SGFW discharge output at https://doi.org/10.17605/OSF. IO/MBR9F.

References

- Adusumilli, S., Fricker, H. A., Medley, B., Padman, L., & Siegfried, M. R. (2020). Interannual variations in meltwater input to the southern ocean from Antarctic ice shelves. *Nature Geoscience*, 13(9), 616–620. https://doi.org/10.1038/s41561-020-0616-z
- Alley, K. E., Scambos, T. A., Siegfried, M. R., & Fricker, H. A. (2016). Impacts of warm water on Antarctic ice shelf stability through basal channel formation. *Nature Geoscience*, 9(4), 290–293. https://doi.org/10.1038/ngeo2675
- Andrés, E. D., Slater, D. A., Straneo, F., Otero, J., Das, S., & Navarro, F. (2020). Surface emergence of glacial plumes determined by fjord stratification. *The Cryosphere*, 14(6), 1951–1969. https://doi.org/10.5194/tc-14-1951-2020
- Cai, C., Rignot, E., Menemenlis, D., & Nakayama, Y. (2017). Observations and modeling of ocean-induced melt beneath Petermann glacier ice shelf in northwestern Greenland. *Geophysical Research Letters*, 44(16), 8396–8403. https://doi.org/10.1002/2017g1073711

Cook, S., Nicholls, K. W., Vaňková, I., Thompson, S. S., & Galton-Fenzi, B. K. (2023). Data initiatives for ocean-driven melt of Antarctic ice shelves. Annals of Glaciology, 1–6. https://doi.org/10.1017/aog.2023.6

- Cook, S. J., Christoffersen, P., Todd, J., Slater, D., & Chauché, N. (2020). Coupled modelling of subglacial hydrology and calving-front melting at store glacier, West Greenland. *The Cryosphere*, 14(3), 905–924. https://doi.org/10.5194/tc-14-905-2020
- Davis, P. E., Nicholls, K. W., Holland, D. M., Schmidt, B. E., Washam, P., Riverman, K. L., et al. (2023). Suppressed basal melting in the eastern Thwaites glacier grounding zone. *Nature*, 614(7948), 479–485. https://doi.org/10.1038/s41586-022-05586-0
- Dee, D. P., Uppala, S. M., Simmons, A., Berrisford, P., Poli, P., Kobayashi, S., et al. (2011). The ERA-interim reanalysis: Configuration and performance of the data assimilation system. *Quarterly Journal of the Royal Meteorological Society*, 137(656), 553–597. https://doi. org/10.1002/qi.828
- Dinniman, M. S., Asay-Davis, X. S., Galton-Fenzi, B. K., Holland, P. R., Jenkins, A., & Timmermann, R. (2016). Modeling ice shelf/ocean interaction in Antarctica: A review. Oceanography, 29(4), 144–153. https://doi.org/10.5670/oceanog.2016.106
- Dinniman, M. S., Klinck, J. M., & Smith, W. O., Jr. (2007). Influence of sea ice cover and icebergs on circulation and water mass formation in a numerical circulation model of the Ross Sea, Antarctica. *Journal of Geophysical Research*, 112(C11), C11013. https://doi. org/10.1029/2006jc004036
- Dow, C., McCormack, F., Young, D., Greenbaum, J., Roberts, J., & Blankenship, D. (2020). Totten glacier subglacial hydrology determined from geophysics and modeling. *Earth and Planetary Science Letters*, 531, 115961. https://doi.org/10.1016/j.epsl.2019.115961
- Dow, C., Ross, N., Jeofry, H., Siu, K., & Siegert, M. (2022). Antarctic basal environment shaped by high-pressure flow through a subglacial river system. *Nature Geoscience*, 15(11), 1–7. https://doi.org/10.1038/s41561-022-01059-1
- Dupont, T. K., & Alley, R. B. (2005). Assessment of the importance of ice-shelf buttressing to ice-sheet flow. *Geophysical Research Letters*, 32(4), 1–4. https://doi.org/10.1029/2004GL022024
- Flament, T., & Rémy, F. (2012). Dynamic thinning of Antarctic glaciers from along-track repeat radar altimetry. *Journal of Glaciology*, 58(211), 830–840. https://doi.org/10.3189/2012jog11j118
- Fricker, H. A., Scambos, T., Bindschadler, R., & Padman, L. (2007). An active subglacial water system in west Antarctica mapped from space. Science, 315(5818), 1544–1548. https://doi.org/10.1126/science.1136897
- Fried, M. J., Catania, G. A., Bartholomaus, T. C., Duncan, D., Davis, M., Stearns, L. A., et al. (2015). Distributed subglacial discharge drives significant submarine melt at a Greenland tidewater glacier. *Geophysical Research Letters*, 42(21), 9328–9336. https://doi.org/10.1002/2015gl065806
- Fürst, J. J., Durand, G., Gillet-Chaulet, F., Tavard, L., Rankl, M., Braun, M., & Gagliardini, O. (2016). The safety band of Antarctic ice shelves. *Nature Climate Change*, 6(5), 479–482. https://doi.org/10.1038/nclimate2912
- Galton-Fenzi, B., Hunter, J., Coleman, R., Marsland, S., & Warner, R. (2012). Modeling the basal melting and marine ice accretion of the Amery ice shelf. *Journal of Geophysical Research*, 117(C9), C09031. https://doi.org/10.1029/2012jc008214
- Gardner, A. S., Moholdt, G., Scambos, T., Fahnstock, M., Ligtenberg, S., van den Broeke, M., & Nilsson, J. (2018). Increased West Antarctic and unchanged east Antarctic ice discharge over the last 7 years. *The Cryosphere*, 12(2), 521–547. https://doi.org/10.5194/tc-12-521-2018
- Gladish, C. V., Holland, D. M., Holland, P. R., & Price, S. F. (2012). Ice-shelf basal channels in a coupled ice/ocean model. *Journal of Glaciology*, 58(212), 1227–1244. https://doi.org/10.3189/2012J0G12J003
- Goldberg, D., Gourmelen, N., Kimura, S., Millan, R., & Snow, K. (2019). How accurately should we model ice shelf melt rates? *Geophysical Research Letters*, 46(1), 189–199. https://doi.org/10.1029/2018gl080383
- Gourmelen, N., Escorihuela, M., Shepherd, A., Foresta, L., Muir, A., Garcia-Mondéjar, A., et al. (2018). CryoSat-2 swath interferometric altimetry for mapping ice elevation and elevation change. *Advances in Space Research*, 62(6), 1226–1242. https://doi.org/10.1016/j.asr.2017.11.014
- Gourmelen, N., Goldberg, D. N., Snow, K., Henley, S. F., Bingham, R. G., Kimura, S., et al. (2017). Channelized melting drives thinning under a rapidly melting Antarctic ice shelf. *Geophysical Research Letters*, 44(19), 9796–9804. https://doi.org/10.1002/2017gl074929
- Greenbaum, J., Blankenship, D., Young, D., Richter, T., Roberts, J., Aitken, A., et al. (2015). Ocean access to a cavity beneath Totten glacier in east Antarctica. *Nature Geoscience*, 8(4), 294–298. https://doi.org/10.1038/ngeo2388

Acknowledgments

We thank Kate Hedstrom and John Wilkin for model configuration advice; Keith Nicholls for illuminating discussions on the thermal impact of subglacial discharge; and, Xylar Asay-Davis and two anonymous reviewers for comments that improved this manuscript. There are no real or perceived financial conflicts of interests for any author. Maps produced with AMT (Greene, Gwyther, & Blankenship, 2017) and CDT (Greene et al., 2019). CD was supported by the Natural Sciences and Engineering Research Council of Canada (NSERC; RGPIN-03761-2017) and the Canada Research Chairs Program (950-231237). NG was supported via the European Space Agency Polar+Ice Shelves project 4000132186/20/I-EF. BG-F received support from the Australian Government as part of the Antarctic Science Collaboration Initiative programme (Project ASCI00002). This research was undertaken with assistance from the National Computational Infrastructure. Open access publishing facilitated by The University of Queensland, as part of the Wiley - The University of Queensland agreement via the Council of Australian University Librarians.

- Greene, C. A., Gwyther, D. E., & Blankenship, D. D. (2017). Antarctic mapping tools for MATLAB. Computers & Geosciences, 104, 151–157. https://doi.org/10.1016/j.cageo.2016.08.003
- Greene, C. A., Blankenship, D. D., Gwyther, D. E., Silvano, A., & van Wijk, E. (2017). Wind causes Totten Ice Shelf melt and acceleration. *Science Advances*, 3(11), e1701681. https://doi.org/10.1126/sciadv.1701681
- Greene, C. A., Thirumalai, K., Kearney, K. A., Delgado, J. M., Schwanghart, W., Wolfenbarger, N. S., et al. (2019). The climate data toolbox for MATLAB. *Geochemistry*, *Geophysics*, *Geosystems*, 20(7), 3774–3781. https://doi.org/10.1029/2019gc008392
- Gwyther, D. E., Galton-Fenzi, B. K., Hunter, J. R., & Roberts, J. L. (2014). Simulated melt rates for the Totten and Dalton ice shelves. Ocean Science, 10(3), 267–279. https://doi.org/10.5194/os-10-267-2014
- Gwyther, D. E., O'Kane, T. J., Galton-Fenzi, B. K., Monselesan, D. P., & Greenbaum, J. S. (2018). Intrinsic processes drive variability in basal melting of the Totten glacier ice shelf. *Nature Communications*, 9(1), 1–8. https://doi.org/10.1038/s41467-018-05618-2
- Hellmer, H. H., & Olbers, D. J. (1989). A two-dimensional model for the thermohaline circulation under an ice shelf. Antarctic Science, 1(4), 325–336. https://doi.org/10.1017/s0954102089000490
- Hewitt, I. J. (2020). Subglacial plumes. Annual Review of Fluid Mechanics, 52(1), 145–169. https://doi.org/10.1146/annurev-fluid-010719-060252
 Holland, D. M., & Jenkins, A. (1999). Modeling thermodynamic ice–ocean interactions at the base of an ice shelf. Journal of Physical Oceanography, 29(8), 1787–1800. https://doi.org/10.1175/1520-0485(1999)029<1787:mtioia>2.0.co;2
- Howat, I. M., Porter, C., Smith, B. E., Noh, M.-J., & Morin, P. (2019). The reference elevation model of Antarctica. *The Cryosphere*, 13(2), 665–674. https://doi.org/10.5194/tc-13-665-2019
- Jenkins, A. (2011). Convection-driven melting near the grounding lines of ice shelves and tidewater glaciers. *Journal of Physical Oceanography*, 41(12), 2279–2294. https://doi.org/10.1175/jpo-d-11-03.1
- Large, W. G., McWilliams, J. C., & Doney, S. C. (1994). Oceanic vertical mixing: A review and a model with a nonlocal boundary layer parameterization. *Reviews of Geophysics*, 32(4), 363–403. https://doi.org/10.1029/94rg01872
- Le Brocq, A. M., Ross, N., Griggs, J. a., Bingham, R. G., Corr, H. F. J., Ferraccioli, F., et al. (2013). Evidence from ice shelves for channelized meltwater flow beneath the Antarctic Ice Sheet. *Nature Geoscience*, 6(10), 1–4. https://doi.org/10.1038/ngeo1977
- Lewis, E., & Perkin, R. (1986). Ice pumps and their rates. Journal of Geophysical Research, 91(C10), 11756–11762. https://doi.org/10.1029/jc091ic10p11756
- Li, X., Rignot, E., Morlighem, M., Mouginot, J., & Scheuchl, B. (2015). Grounding line retreat of Totten glacier, east Antarctica, 1996 to 2013. Geophysical Research Letters, 42(19), 8049–8056. https://doi.org/10.1002/2015gl065701
- Marsh, O. J., Fricker, H. A., Siegfried, M. R., Christianson, K., Nicholls, K. W., Corr, H. F., & Catania, G. (2016). High basal melting forming a channel at the grounding line of Ross Ice Shelf, Antarctica. *Geophysical Research Letters*, 43(1), 250–255. https://doi.org/10.1002/2015gl066612
- Marshall, J., Hill, C., Perelman, L., & Adcroft, A. (1997). Hydrostatic, quasi-hydrostatic, and nonhydrostatic ocean modeling. Journal of Geophysical Research, 102(C3), 5733–5752. https://doi.org/10.1029/96JC02776
- McPhee, M. G., Maykut, G. A., & Morison, J. H. (1987). Dynamics and thermodynamics of the ice/upper ocean system in the marginal ice zone of the Greenland Sea. Journal of Geophysical Research, 92(C7), 7017–7031. https://doi.org/10.1029/jc092ic07p07017
- Menemenlis, D., Campin, J.-M., Heimbach, P., Hill, C., Lee, T., Nguyen, A., et al. (2008). ECCO2: High resolution global ocean and sea ice data synthesis. *Mercator Ocean Quarterly Newsletter*, 31(October), 13–21.
- Motyka, R. J., Hunter, L., Echelmeyer, K. A., & Connor, C. (2003). Submarine melting at the terminus of a temperate tidewater glacier, Leconte glacier, Alaska, USA. Annals of Glaciology, 36, 57–65. https://doi.org/10.3189/172756403781816374
- Nakayama, Y., Cai, C., & Seroussi, H. (2021). Impact of subglacial freshwater discharge on pine island ice shelf. *Geophysical Research Letters*, 48(18), e2021GL093923. https://doi.org/10.1029/2021GL093923
- Pritchard, H., Ligtenberg, S. R., Fricker, H. A., Vaughan, D. G., van den Broeke, M. R., & Padman, L. (2012). Antarctic ice-sheet loss driven by basal melting of ice shelves. *Nature*, 484(7395), 502–505. https://doi.org/10.1038/nature10968
- Rintoul, S. R., Silvano, A., Pena-Molino, B., van Wijk, E., Rosenberg, M., Greenbaum, J. S., & Blankenship, D. D. (2016). Ocean heat drives rapid basal melt of the Totten Ice Shelf. Science Advances, 2(12), e1601610. https://doi.org/10.1126/sciadv.1601610
- Roberts, J., Galton-Fenzi, B. K., Paolo, F. S., Donnelly, C., Gwyther, D. E., Padman, L., et al. (2018). Ocean forced variability of Totten glacier mass loss. *Geological Society, London, Special Publications*, 461(1), 175–186. https://doi.org/10.1144/sp461.6
- Scambos, T. A., Bohlander, J., Shuman, C. A., & Skvarca, P. (2004). Glacier acceleration and thinning after ice shelf collapse in the Larsen b embayment, Antarctica. *Geophysical Research Letters*, 31(18), L18402. https://doi.org/10.1029/2004gl020670
- Scambos, T. A., Haran, T. M., Fahnestock, M., Painter, T., & Bohlander, J. (2007). Modis-based mosaic of Antarctica (MOA) data sets: Continent-wide surface morphology and snow grain size. *Remote Sensing of Environment*, 111(2–3), 242–257. https://doi.org/10.1016/j. rse.2006.12.020
- Shchepetkin, A. F., & McWilliams, J. C. (2005). The regional oceanic modeling system (ROMS): A split-explicit, free-surface, topography-following-coordinate oceanic model. *Ocean Modelling*, 9(4), 347–404. https://doi.org/10.1016/j.ocemod.2004.08.002
- Silvano, A., Rintoul, S. R., Kusahara, K., Peña-Molino, B., van Wijk, E., Gwyther, D. E., & Williams, G. D. (2019). Seasonality of warm water intrusions onto the continental shelf near the Totten glacier. *Journal of Geophysical Research: Oceans*, 124(6), 4272–4289. https://doi.org/10.1029/2018jc014634
- Silvano, A., Rintoul, S. R., Peña-Molino, B., & Williams, G. D. (2017). Distribution of water masses and meltwater on the continental shelf near the Totten and Moscow University ice shelves. *Journal of Geophysical Research: Oceans*, 122(3), 2050–2068. https://doi. org/10.1002/2016jc012115
- Slater, D., Nienow, P., Cowton, T., Goldberg, D., & Sole, A. (2015). Effect of near-terminus subglacial hydrology on tidewater glacier submarine melt rates. *Geophysical Research Letters*, 42(8), 2861–2868. https://doi.org/10.1002/2014gl062494
- Slater, D., Straneo, F., Das, S., Richards, C., Wagner, T., & Nienow, P. (2018). Localized plumes drive front-wide Ocean melting of a Greenlandic tidewater glacier. *Geophysical Research Letters*, 45(22), 12–350. https://doi.org/10.1029/2018gl080763
- Smith, B., Fricker, H. A., Gardner, A. S., Medley, B., Nilsson, J., Paolo, F. S., et al. (2020). Pervasive ice sheet mass loss reflects competing ocean and atmosphere processes. *Science*, 368(6496), 1239–1242. https://doi.org/10.1126/science.aaz5845
- Tamura, T., Ohshima, K. I., Fraser, A. D., & Williams, G. D. (2016). Sea ice production variability in Antarctic coastal polynyas. Journal of Geophysical Research: Oceans, 121(5), 2967–2979. https://doi.org/10.1002/2015jc011537
- Tamura, T. P., Nomura, D., Hirano, D., Tamura, T., Kiuchi, M., Hashida, G., et al. (2022). Impacts of basal melting of the Totten Ice Shelf and biological productivity on marine biogeochemical components in Sabrina coast, East Antarctica. *Global Biogeochemical Cycles*, 36, e2022GB007510. https://doi.org/10.1029/2022GB007510
- Tsai, V. C., Stewart, A. L., & Thompson, A. F. (2015). Marine ice-sheet profiles and stability under Coulomb basal conditions. Journal of Glaciology, 61(226), 205–215. https://doi.org/10.3189/2015jog14j221

- Van Achter, G., Fichefet, T., Goosse, H., & Moreno-Chamarro, E. (2022). Influence of fast ice on future ice shelf melting in the Totten glacier area, east Antarctica. *The Cryosphere*, 16(11), 4745–4761. https://doi.org/10.5194/tc-16-4745-2022
- Vaňková, I., Cook, S., Winberry, J. P., Nicholls, K. W., & Galton-Fenzi, B. K. (2021). Deriving melt rates at a complex ice shelf base using in situ radar: Application to Totten Ice Shelf. *Geophysical Research Letters*, 48(7), e2021GL092692. https://doi.org/10.1029/2021gl092692
- Vaňková, I., & Nicholls, K. W. (2022). Ocean variability beneath the filchner-ronne ice shelf inferred from basal melt rate time series. Journal of Geophysical Research: Oceans, 127(10), e2022JC018879. https://doi.org/10.1029/2022JC018879
- Velicogna, I., Mohajerani, Y., Landerer, F., Mouginot, J., Noel, B., Rignot, E., et al. (2020). Continuity of ice sheet mass loss in Greenland and Antarctica from the grace and grace follow-on missions. *Geophysical Research Letters*, 47(8), e2020GL087291. https://doi. org/10.1029/2020GL087291
- Wåhlin, A. K., Yuan, X., Björk, G., & Nohr, C. (2010). Inflow of warm circumpolar deep water in the central Amundsen shelf. Journal of Physical Oceanography, 40(6), 1427–1434. https://doi.org/10.1175/2010JPO4431.1
- Warner, R. C., Fricker, H. A., Adusumilli, S., Arndt, P., Kingslake, J., & Spergel, J. (2021). Rapid formation of an ice Doline on Amery ice shelf, east Antarctica. *Geophysical Research Letters*, 48, e2020GL091095. https://doi.org/10.1029/2020GL091095
- Weertman, J. (1974). Stability of the junction of an ice sheet and an ice shelf. Journal of Glaciology, 13(67), 3–11. https://doi.org/10.3189/ s0022143000023327
- Wei, W., Blankenship, D. D., Greenbaum, J. S., Gourmelen, N., Dow, C. F., Richter, T. G., et al. (2020). Getz ice shelf melt enhanced by freshwater discharge from beneath the west Antarctic ice sheet. *The Cryosphere*, 14(4), 1399–1408. https://doi.org/10.5194/tc-14-1399-2020
- Werder, M. A., Hewitt, I. J., Schoof, C. G., & Flowers, G. E. (2013). Modeling channelized and distributed subglacial drainage in two dimensions. Journal of Geophysical Research: Earth Surface, 118(4), 2140–2158. https://doi.org/10.1002/jgrf.20146
- Wessem, J. M. v., Berg, W. J. v. d., Noël, B. P., Meijgaard, E. v., Amory, C., Birnbaum, G., et al. (2018). Modelling the climate and surface mass balance of polar ice sheets using RACMO2–part 2: Antarctica (1979–2016). *The Cryosphere*, 12(4), 1479–1498. https://doi.org/10.5194/ tc-12-1479-2018
- Williams, G., Meijers, A., Poole, A., Mathiot, P., Tamura, T., & Klocker, A. (2011). Late winter oceanography off the Sabrina and Banzare coast (117–128 E), East Antarctica. Deep Sea Research Part II: Topical Studies in Oceanography, 58(9–10), 1194–1210. https://doi.org/10.1016/j. dsr2.2010.10.035
- Xia, Y., Gwyther, D. E., Galton-Fenzi, B., Cougnon, E. A., Fraser, A. D., & Moore, J. C. (2023). Eddy and tidal driven basal melting of the Totten and Moscow University ice shelves. *Frontiers in Marine Science*, 10. https://doi.org/10.3389/fmars.2023.1159353
- Zhang, W. G., Wilkin, J. L., & Chant, R. J. (2009). Modeling the pathways and mean dynamics of river plume dispersal in the New York bight. Journal of Physical Oceanography, 39(5), 1167–1183. https://doi.org/10.1175/2008jpo4082.1